Author's response to reviews

Title: Virtual House Calls for Parkinson disease (Connect.Parkinson): study protocol for a randomized, controlled trial

Authors:

Meredith A Achey (meredith.achey@chet.rochester.edu)
Christopher A Beck (beck@bst.rochester.edu)
Denise B Beran (dberan@parkinson.org)
Cynthia M Boyd (cyboyd@jhmi.edu)
Peter N Schmidt (pschmidt@parkinson.org)
Allison W Willis (Allison.Willis@uphs.upenn.edu)
Sara S Riggare (sara@riggare.se)
Richard B Simone (rbsimone@pacbell.net)
Kevin M Biglan (kevin_biglan@urmc.rochester.edu)
E. Ray Dorsey (ray.dorsey@chet.rochester.edu)

Version: 3
Date: 4 November 2014

Author's response to reviews: see over
November 4, 2014

Dear Trials Editorial team,

Thank you for your email regarding the changes necessary to our manuscript (MS: 1203586101136568). We have addressed the issues raised in the following ways:

- Please include the names of all ethical bodies that approved your study in the various centres involved. If you do not wish to list them all in the methods section, please include the list as an additional file and refer to this in the methods section.

This documentation has been added as Appendix 1.

Furthermore, we have changed Figure 1 to better illustrate the disparities in access to neurologic care in the U.S., and reference this figure earlier in the text.

In addition, we have made some editorial changes for clarity, updated authors’ institutional contact information, and deleted the former Figure 1 and “ConnectParkinson Study Site Status.xlsx” (now covered in Appendix 1).

Thank you for your acceptance of this manuscript.

Sincerely,

Cynthia M. Boyd, MD, MPH